Author's response to reviews

Title: Human Physiologically Based Pharmacokinetic Model for Propofol

Authors:

David G Levitt (levit001@umn.edu)
Thomas W Schnider (thomas.schnider@kssg.ch)

Version: 7 Date: 30 March 2005

Author's response to reviews:

This revision is nearly identical to the last submission, which represented the final version of the accepted paper. The only change is that some figures have been cropped to decrease the white space.